Author's response to reviews

Title: A simple way to compare the efficiency of outreach chlamydia screening activities

Authors:

Francis J Bowden (frank.bowden@act.gov.au)
Marian J Currie (marian.currie@act.gov.au)
Maureen Todkill (maurenn.todkill@act.gov.au)
Mathias Schmidt (gestalt@mac.com)
Sue Webeck (S.Webeck@shfpact.org.au)
Rendry Del Rosario (Rendry.DelRosario@act.gov.au)
Tim Bavinton (tbavinton@shfpact.org.au)
Alexandra Tyson (Alexandra.Tyson@act.gov.au)

Version: 2 Date: 4 October 2011

Author's response to reviews:

4 October 2011

Dear Editors

Re: A simple way to compare the efficiency of outreach chlamydia screening activities


Thank you for re-considering this manuscript for publication in the BMC Public Health and apologise for the delay in replying.

We have made the requested changes to the manuscript by including the context information in the Background section of the Abstract and included an ethics statement in the Methods section.

The manuscript has been seen and re-approved by all authors and has not been simultaneously submitted or published elsewhere. We look forward to your decision.

Yours sincerely

Frank Bowden
Professor of Medicine